Author's response to reviews

Title: Self-care coping strategies in people with diabetes: a qualitative exploratory study

Authors:

Margaret M Collins (MCollins@co.tuolumne.ca.us)
Colin P Bradley (c.bradley@ucc.ie)
Tony O'Sullivan (tonyosu@eircom.net)
Ivan J Perry (i.perry@ucc.ie)

Version: 4 Date: 6 February 2009

Author's response to reviews:

Dear Editors,

Thank you for accepting our manuscript to be published in your journal. We have ammended the formatting changes to our manuscript, BMC Endocrine Disorders MS : 155539498193412 - Self-care coping strategies in people with diabetes: a qualitative exploratory study, as you have suggested. Thank you for your help.

Regards,

Ivan Perry.